Author’s response to reviews

Title: Rotation of a toric intraocular lens with and without capsular tension ring: data from a multicenter non-inferiority randomized clinical trial (RCT)

Authors:

Ursula Hahn (Ursula.Hahn@uni-wh.de)
Frank Krummenauer (Frank.Krummenauer@uni-wh.de)
Stefanie Schmickler (st.schmickler@augen-zentrum-nordwest.de)
Jörg Koch (joerg.koch@augen-franziskus.de)

Version: 4 Date: 08 Jun 2019

Author’s response to reviews:

Revision Notes

Editor Comments:

1. Consent for publication
We note that you have not included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Authors: We have added a „Consent for publication“ section and stated „Not Applicable“

2. Trial registration
I apologize for the conflicting comments given to you during the final revisions process, however, we would like to ask if you could put the trial information that you were previously asked to remove back into your manuscript. Please see below for exact details.
The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section.
Please provide the date of registration for your trial in the Trial registration section. If the trial was registered retrospectively (registered after the study had already started) please state “Retrospectively registered” at the end of this section as well.
Authors: We have added the requested information as the last paragraph of the abstract.

3. Author contributions
We note the author JK has not confirmed that they have approved the final manuscript. Please provide a statement confirming this author also approved the final manuscript.

Authors: All authors, including Jörg Koch (JK), approved the final manuscript.

5. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please remove any files that should not be published along with your manuscript e.g. cover letters, reviewer responses, guidelines etc., ready for publication.

Authors: We uploaded a single clean copy of the manuscript.